Author’s response to reviews

Title: Barriers and enablers in the management of tuberculosis treatment in Addis Ababa, Ethiopia: A qualitative study

Authors:

Mette Sagbakken (mette.sagbakken@medisin.uio.no)
Jan C. Frich (j.c.d.frich@medisin.uio.no)
Gunnar Bjune (g.a.bjune@medisin.uio.no)

Version: 3 Date: 7 January 2008

Author’s response to reviews:

Dear Editor,

Thank you for the feedback on the manuscript “Barriers and enablers in the management of tuberculosis treatment in Addis Ababa, Ethiopia: A qualitative study” (7086054501421678).

We have now made the requested changes related to the format. The list of authors in the manuscript is written in the same way as in the submission system, and “Reference list” is changed to “References”. Table 3 is referred to under “individual in-depth interviews”, page 8. Due to the order of table referring in the text the earlier “table 3” is changed to “table 2”, and what was “table 2” is changed to “table 3”. We have also changed the minor language errors pointed to by reviewer 3 in page 7, 12 and 26.

We hope the revisions are satisfactory.

With regards

Mette Sagbakken, corresponding author.